<%namespace name="tools" file="tools.tex"/>

\section{Large scale laboratory tests with revetments}

\subsection{M1797: Deltaflume 1981}

\autoref{fig:Deltaflume_M1797_T02_final_profiles} depicts the Schouwen profile from the M1797 test with dune revetment\citep{Vellinga1981b}.

${tools.image('Deltaflume_M1797/T02/', 'final_profiles.png', 'Final profile of test T02')}

\subsection{H298: Deltaflume 1986}

\citet{Steetzel1987} describes a series of large scale experiments with revetments of different heights in the Delta Flume. A depthscale $n_d$ = 5 is used for all experiments \citep{Vellinga1986} and the initial profile in the flume correponds to the reference profile for the Holland coast. The location of the top of the revetment varied in each experiment, as can be seen in \autoref{fig:DeltaflumeH298_T1_final_profiles} to \autoref{fig:DeltaflumeH298_T3_final_profiles}.

${tools.image('DeltaflumeH298/T1/', 'final_profiles.png', 'Final profile of test 1')}
${tools.image('DeltaflumeH298/T2/', 'final_profiles.png', 'Final profile of test 2')}
${tools.image('DeltaflumeH298/T3/', 'final_profiles.png', 'Final profile of test 3')}

\subsection{H4731: Deltaflume 2006}

In 2006, Delta Flume large scale experiments (depth scale of 6) were performed to gage the influence of collapsing dune revetments on dune erosion\citep{VanGent2007}. Since XBeach can only handle solid structures, the collapse of the structure is not included in the simulations. \autoref{fig:Deltaflume_H4731_T11_final_profiles} to \autoref{fig:Deltaflume_H4731_T12_final_profiles} show the last measurement before the start of the collapse of the structure.

${tools.image('Deltaflume_H4731/T11/', 'final_profiles.png', 'Final profile of test 11')}
${tools.image('Deltaflume_H4731/T12/', 'final_profiles.png', 'Final profile of test 12')}